Author’s response to reviews

Title: Undoing the unspeakable: researching racism in Swedish healthcare using a participatory process to build dialogue

Authors:

Hannah Bradby (hannah.bradby@soc.uu.se)

Suruchi Thapar-Bjorket (suruchi.thapar-bjorkert@statsvet.uu.se)

Sarah Hamed (sarah.hamed@soc.uu.se)

Beth Ahlberg (beth.ahlberg@vgregion.se)

Version: 1 Date: 04 Mar 2019

Author’s response to reviews:

Dear Editor

The decision letter on 11th Feb requested that we include <<more detail in the methods section>> to give a <<clearer picture of the methods that will be used at each phase or, where these will be developed as the project is undertaken, how this will be managed.>>

Pages 11, 12, 13 and 14 have been substantively revised, to supply a fuller picture of the planning for this research study and to make clear the moments where it is difficult to anticipate stages due to the participatory input.

Yours faithfully, Hannah Bradby on behalf of the authors.